Author’s response to reviews

Title: Polymethylmethacrylate-assisted Ventral Discectomy: Rate of Pseudarthrosis and Clinical Outcome with a Minimum Follow-up of 5 Years

Authors:

Mario Cabraja (mario.cabraja@charite.de)
Daniel Koeppen (dani_san@gmx.de)
Wolfgang R Lanksch (lanksch@ini-hannover.de)
Klaus Maier-Hauff (klausmaierhauff@yahoo.de)
Stefan N Kroppenstedt (stefan.kroppenstedt@charite.de)

Version: 2 Date: 10 December 2010

Author’s response to reviews:

We have added the following paragraph into the method section (second block):

"The research conforms to the Helsinki Declaration, and to local legislation. By ensuring the patient anonymity, an approval of our institutional ethics committee is not required for a retrospective study."